Author's response to reviews

Title: A Case-Control Study of Occupational Magnetic Field Exposure and Alzheimer's Disease: Results from the California Alzheimer's Disease and Treatment Centers

Authors:

Zoreh Davanipour (zdavani@aol.com)
Chiu-Chen Tseng (annchiuchen@yahoo.com)
Pey-Juan Lee (peylee@usc.edu)
Eugene Sobel (sobel55@earthlink.net)

Version: 4 Date: 16 April 2007

Author's response to reviews: see over
Response to Reviewer #4's Report: 13 February 2007

We thank Dr. Ward for his careful reviews.

1. The discussion in the text (p. 11) related to Table 3 has been corrected:
   a. The p-value was 0.002, as provided in Table 3;
   b. The mean differences in age-at-diagnosis and age-at-symptom onset were statistically significant, as stated in Table 3;
   c. The number of ordinal categories for age-at-diagnosis and age-at-symptom onset in Table 3 is correct, and the wording in the text has been corrected.

2. With respect to the statement about seamstresses in #3 on Page 18, we have provided the Mantel-Haenszel odds ratio for combined case-control studies from the three (3) papers. We have not provided the actual data from the two Sobel et al. papers, which are available in the papers themselves. We could provide the detailed data in a table if the editors want us to do so.